



Extended Research Article

Effectiveness of biomarker-guided duration of antibiotic treatment in children hospitalised with confirmed or suspected bacterial infection: the BATCH RCT

Cherry-Ann Waldron,^{1*} Philip Pallmann,¹ Simon Schoenbuchner,² Debbie Harris,¹ Lucy Brookes-Howell,¹ Céu Mateus,³ Jolanta Bernatoniene,⁴ Katrina Cathie,^{5,6,7} Saul N Faust,^{5,6,7} Josie Henley,⁸ Lucy Hinds,⁹ Kerry Hood,¹⁰ Chao Huang,¹¹ Sarah Jones,¹² Sarah Kotecha,¹ Sarah Milosevic,¹ Helen Nabwera,^{13,14,15} Sanjay Patel,^{5,7} Stéphane Paulus,^{16,17} Colin VE Powell,² Jenny Preston,¹⁸ Huasheng Xiang,^{19,20} Emma Thomas-Jones^{1†} and Enitan D Carrol^{15,21†} on behalf of the BATCH trial team

*Corresponding author waldronc@cardiff.ac.uk

¹Centre for Trials Research, Cardiff University, Neuadd Meirionnydd, Heath Park, Cardiff, UK

²Division of Population Medicine, School of Medicine, Cardiff University, Heath Park, Cardiff, UK

³Division of Health Research, Lancaster University, Lancaster, UK

⁴University Hospitals Bristol and Weston NHS Foundation Trust, Bristol Royal Hospital for Children, Bristol, UK

⁵NIHR Southampton Clinical Research Facility and Biomedical Research Centre, University Hospital Southampton NHS Foundation Trust, Southampton, UK

⁶Faculty of Medicine and Institute for Life Sciences, University of Southampton, Southampton, UK

⁷Department of Paediatric Infectious Diseases, University Hospital Southampton NHS Foundation Trust, Southampton, UK

⁸School of Social Sciences, Cardiff University, Cardiff, UK

⁹Sheffield Children's NHS Foundation Trust, Sheffield, UK

¹⁰College of Biomedical Life Sciences, Cardiff University, Cardiff, UK

¹¹Hull York Medical School, University of Hull, Hull, UK

¹²Involve Network, Health and Care Research Wales, Cardiff, UK

¹³Liverpool School of Tropical Medicine, Pembroke Place, Liverpool, UK

¹⁴Centre of Excellence in Women and Child Health, Aga Khan University, Nairobi, Kenya

¹⁵Department of Infectious Diseases, Alder Hey Children's NHS Foundation Trust, Liverpool, UK

¹⁶Oxford University Hospitals NHS Foundation Trust, Oxford, UK

¹⁷Department of Paediatrics, University of Oxford Oxford, UK

¹⁸NIHR Alder Hey Clinical Research Facility, Alder Hey Children's NHS Foundation Trust, Liverpool, UK

¹⁹Management School, Lancaster University, Lancaster, UK

²⁰Lancaster University College at Beijing Jiaotong University, Shandong, China

²¹Department of Clinical Infection, Microbiology and Immunology, Institute of Infection, Veterinary and Ecological Sciences, University of Liverpool, Liverpool, UK

[†]Joint last authors

Published May 2025 DOI: 10.3310/MBVA3675

Plain language summary

Effectiveness of biomarker-guided duration of antibiotic treatment in children hospitalised with confirmed or suspected bacterial infection: the BATCH RCT

Health Technology Assessment 2025; Vol. 29: No. 16

DOI: 10.3310/MBVA3675

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

Adaily task in hospitals is to assess whether sick children have an infection or not, and doctors need to decide whether to start, stop or change antibiotics. On one hand, giving antibiotics promptly saves lives, but on the other, giving antibiotics to people who do not need them leads to overuse of antibiotics resulting in antibiotics no longer working for infections, so-called antibiotic resistance. If we can reduce antibiotic use in hospitals, this would be an important step in combating the spread of hospital superbugs.

Blood tests can be used to measure the body's response to infection. Most hospitals in the National Health Service use blood tests to monitor whether a person is responding to antibiotics. One example is C-reactive protein, but this test does not always tell you whether there is an infection there and if it is getting better, or whether the person is just unwell from another reason. A blood test measuring procalcitonin is better for diagnosing bacterial infections, and procalcitonin levels are quicker to decrease when a patient starts to improve and antibiotics start working, compared to C-reactive protein levels. However, procalcitonin tests are not routinely used for children in the National Health Service.

The BATCH trial looked at whether the use of a procalcitonin test is safe and could help doctors decide whether to stop or change antibiotics (from intravenous to oral), both of which safely reduce antibiotic use (and help limit antibiotic resistance), compared to not using the test.

The trial found that in children admitted to hospital with a bacterial infection, the addition of the procalcitonin test is safe to use but does not reduce how long intravenous antibiotics were given for.

Doctors did not always use the procalcitonin result when making antibiotic decisions, and although parents were largely positive about participation in the trial, some had concerns about extra blood tests and clinicians stopping antibiotics too early. Future research should include education and training for doctors to ensure that the procalcitonin test forms part of routine care.

Health Technology Assessment

ISSN 2046-4924 (Online)

Impact factor: 3.5

A list of Journals Library editors can be found on the NIHR Journals Library website

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.5 and is ranked 30th (out of 174 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2022 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), EMBASE (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Manuscripts are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This article

The research reported in this issue of the journal was funded by the HTA programme as award number 15/188/42. The contractual start date was in September 2017. The draft manuscript began editorial review in February 2024 and was accepted for publication in October 2024. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' manuscript and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this article.

This article presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

This article was published based on current knowledge at the time and date of publication. NIHR is committed to being inclusive and will continually monitor best practice and guidance in relation to terminology and language to ensure that we remain relevant to our stakeholders.

Copyright © 2025 Waldron *et al.* This work was produced by Waldron *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).